Author's response to reviews

Title: Structural Relationships among Nurses' Job Rotation, Role Stress, Job Satisfaction, and Organizational Commitment

Authors:

Hsin Hsin Chang (easyhhc@mail.ncku.edu.tw)
Ching Sheng Chang (arthur0660@yahoo.com.tw)

Version: 6 Date: 14 April 2008

Author's response to reviews: see over
BMC Health Services Research
1st Revision: 15 Apr. 2008

Title: Structural Relationships among Nurses’ Job Rotation, Role Stress, Job Satisfaction, and Organizational Commitment
Authors: Hsin Hsin Chang and Ching Sheng Chang
Contact: Ching Sheng Chang
Affiliation: Dept. of Business Administration, National Cheng Kung University
E-mail: arthur0660@yahoo.com.tw

I would like to thank the review panel for making their time to review my submission, and for the indefatigable guidance; I was greatly benefited. I revised manuscript addresses all of the editor and reviewer comments and consulted an English language specialist for help with the language and grammar. Please check! Thank your very much!

Author's response to reviews (1712625865157250 _comment):
1. Had revised in Table 1.
2. Had revised in Table 1 and Literature Review.
3. Had revised in Table 1.
4. Had revised in Table 1.
5. Had revised in page 5.
6. Had revised in Table 1.
7. Had revised in page 10 and elsewhere.
8. Had consulted an English language specialist for help with the language and grammar.
10. Had revised in Table 1, page 11, and page 12.
13. Had revised in Table 3 and Table 4.
14. Had revised in Table 3 and Table 4.
15. Had revised in Table 1.
16. Had consulted an English language specialist for help with the language and grammar.
17. Had revised for each table and figure.
18. Had examined.

Author's response to reviews (1856888492166868 _comment):
1. Had revised in Table 1.
2. Had revised in Table 1.
3. Had revised in Table 1 and page 11.
4. Had revised in Figure 1.
5. Had revised in Figure 1 and Literature Review.
6. Had revised in the literature review and discussion.
7. Had revised.
8. Had revised.